

## 

**Citation:** Partridge FA, Murphy EA, Willis NJ, Bataille CJR, Forman R, Heyer-Chauhan N, et al. (2017) Dihydrobenz[*e*][1,4]oxazepin-2(3*H*)-ones, a new anthelmintic chemotype immobilising whipworm and reducing infectivity *in vivo*. PLoS Negl Trop Dis 11(2): e0005359. doi:10.1371/ journal.pntd.0005359

Editor: Ramesh Ratnappan, George Washington University, UNITED STATES

Received: October 13, 2016

Accepted: January 26, 2017

Published: February 9, 2017

**Copyright:** © 2017 Partridge et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files. Assay data and structures for all 192 screened dihydrobenzoxazepinones are contained in the S1 Dataset. Assay data and structures for the 5 resynthesized confirmed active compounds have been uploaded to the public database PubChem. Accession numbers (SID) are 318018055, 318018056, 318018057, 318018058, and 318018059 and (AID) 1224898. RESEARCH ARTICLE

# Dihydrobenz[*e*][1,4]oxazepin-2(3*H*)-ones, a new anthelmintic chemotype immobilising whipworm and reducing infectivity *in vivo*

Frederick A. Partridge<sup>1</sup>, Emma A. Murphy<sup>2</sup>, Nicky J. Willis<sup>3</sup>, Carole J. R. Bataille<sup>3</sup>, Ruth Forman<sup>2</sup>, Narinder Heyer-Chauhan<sup>1</sup>, Bruno Marinič<sup>3</sup>, Daniel J. C. Sowood<sup>3</sup>, Graham M. Wynne<sup>3</sup>, Kathryn J. Else<sup>2</sup>, Angela J. Russell<sup>3,4</sup>, David B. Sattelle<sup>1</sup>

1 Centre for Respiratory Biology, UCL Respiratory, Division of Medicine, University College London, London, United Kingdom, 2 Faculty of Biology, Medicine and Health, University of Manchester, Manchester, United Kingdom, 3 Department of Chemistry, Chemistry Research Laboratory, University of Oxford, Oxford, United Kingdom, 4 Department of Pharmacology, University of Oxford, Oxford, United Kingdom

• These authors contributed equally to this work.

- ¤ Current address: Liverpool School of Tropical Medicine, Liverpool, United Kingdom
- \* kathryn.else@manchester.ac.uk (KJE); angela.russell@chem.ox.ac.uk (AJR); d.sattelle@ucl.ac.uk (DBS)

### Abstract

Trichuris trichiura is a human parasitic whipworm infecting around 500 million people globally, damaging the physical growth and educational performance of those infected. Current drug treatment options are limited and lack efficacy against the worm, preventing an eradication programme. It is therefore important to develop new treatments for trichuriasis. Using Trichuris muris, an established model for T. trichiura, we screened a library of 480 novel drug-like small molecules for compounds causing paralysis of the ex vivo adult parasite. We identified a class of dihydrobenz[e][1,4]oxazepin-2(3H)-one compounds with anthelmintic activity against T. muris. Further screening of structurally related compounds and resynthesis of the most potent molecules led to the identification of 20 active dihydrobenzoxazepinones, a class of molecule not previously implicated in nematode control. The most active immobilise adult T. muris with EC<sub>50</sub> values around 25–50µM, comparable to the existing anthelmintic levamisole. The best compounds from this chemotype show low cytotoxicity against murine gut epithelial cells, demonstrating selectivity for the parasite. Developing a novel oral pharmaceutical treatment for a neglected disease and deploying it via mass drug administration is challenging. Interestingly, the dihydrobenzoxazepinone OX02983 reduces the ability of embryonated T. muris eggs to establish infection in the mouse host in vivo. Complementing the potential development of dihydrobenzoxazepinones as an oral anthelmintic, this supports an alternative strategy of developing a therapeutic that acts in the environment, perhaps via a spray, to interrupt the parasite lifecycle. Together these results show that the dihydrobenzoxazepinones are a new class of anthelmintic, active against both egg and adult stages of Trichuris parasites. They demonstrate encouraging selectivity for the parasite, and importantly show considerable scope for further optimisation to improve potency and pharmacokinetic properties with the aim of developing a clinical agent.

Funding: F.A.P. is supported by Medical Research Council (www.mrc.ac.uk) grant MR/N024842/1 to D.B.S. and prior to that was supported by CeBioscience Ltd (http://www.cebioscience.com/) and a University College London Fellowship (www. ucl.ac.uk). E.A.M. was supported by a Biotechnology and Biological Sciences Research Council (www.bbsrc.ac.uk) PhD studentship and a Burroughs Wellcome Travel Award (www.bwfund. org). R.F. is supported by Medical Research Council (www.mrc.ac.uk) grant MR/N022661/1 to K.J.E. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

#### Author summary

Trichuris trichiura is a human parasitic whipworm infecting around 500 million people globally and having major consequences on the physical growth and educational performance of those infected. Current drug treatment options are limited and lack efficacy against the worm. Critically, they lack the effectiveness that would allow for a practical program for eradication of this parasite. It is therefore important to develop new treatments for trichuriasis. We screened for molecules that could paralyse the adult of a closely related mouse parasite, and identified a class of compounds, the dihydrobenzoxazepinones, not previously implicated as anthelmintics. Importantly, our compounds are active against the parasite but show only low toxicity against mouse cells, demonstrating selectivity for the parasite. Dihydrobenzoxazepinones could be developed as potential pharmaceutical treatments for trichuriasis. Since developing and deploying new drugs for neglected diseases by mass administration is challenging, we also explored whether the compounds could potentially be used to interrupt the *Trichuris* lifecycle by acting on eggs. Our dihydrobenzoxazepinone compounds reduced the ability of T. muris eggs to establish infection in their mouse host. This supports an environmental spray strategy for the control of Trichuris targeting their eggs in environmental hotspots such as latrines.

#### Introduction

Gastrointestinal nematode infections result in morbidity in humans and livestock; approximately 1 billion people worldwide are infected [1]. Trichuriasis, a neglected human tropical disease caused by infection by the whipworm Trichuris trichiura, affects approximately 500 million people mainly in developing countries, resulting in disability and poor child development [2]. Current strategies for helminth control include Mass Drug Administration (MDA) programs during childhood [2–5]. MDA typically uses benzimidazole anthelmintics against Trichuris, such as albendazole and mebendazole, but both show poor efficacy, with low cure rates and increasing drug resistance; similar problems are encountered when these compounds are used to target livestock Trichuris species [2,6]. Benzimidazoles inhibit microtubule synthesis, normally resulting in parasite immobilization [7] but unlike luminal dwelling nematodes such as Ascaris, where annual treatment with albendazole or mebendazole has resulted in cure rates of up to 95% [8], Trichuris is less well controlled, possibly due to the burrowing of the whipworm into intestinal epithelial cells, which may provide some protection from drugs. Modelling studies have concluded that MDA with benzimidazoles is unable to interrupt transmission of *T. trichiura* and therefore fails to achieve elimination in many settings [9]. Coadministration of two or more anthelmintics, such as albendazole plus ivermectin [10] or albendazole plus oxantel pamoate [10,11] has been demonstrated in randomised controlled trials to improve single-dose cure rates, and a recent study found that oxantel pamoate alone gave a single-dose cure rate of 60% at the optimal dose [12]. However even these improved cure rates are much lower than those achieved for other intestinal nematodes and again modelling suggests that they are insufficient to achieve elimination in higher transmission settings [9]. There are also concerns about the emergence of resistance [8,13]. No vaccine currently exists for trichuriasis. Elimination of this disease will likely require a three-way approach of new, more effective drugs, vaccine development, and complementary interventions in hygiene, sanitation and education.

Given the need for new classes of anti-*Trichuris* compounds, there have been a number of assays developed to screen compounds for activity against this parasite. These include manual

scoring of motility; indirect assessment of viability by using the xCELLigence System; assessment of metabolic activity via colorimetric assays such as resazurin, MTT, and acid phosphatase activity; and assessment of motor activity via isothermal microcalorimetry [14–17].

A variety of plant materials have been shown to have activity against *Trichuris*. Koné and colleagues tested a large panel of plant extracts that are used in the Côte d'Ivoire to treat parasitic diseases for their activity against *T. muris* [18]. Notably, an extract from the leaves of *Combretum mucronatum* both blocked the motility of *T. muris* adults *in vitro* and was effective at reducing worm burden when used to treat mice infected with *T. muris*. Papaya latex (which is rich in cysteine proteases) or the purified proteases ficin and papain rapidly reduce motility of *T. muris* parasites *in vitro*, and this is associated with damage to nematode cuticle [19]. Indeed papaya latex extract is also effective at reducing the worm burden of mice infected with *T. muris*. These cysteine proteases from papaya have also been demonstrated in randomized control trials to be more effective than albendazole at treating *T. suis* in pigs [20]. However the large doses required for efficacy, of the order of 10-40g of protease, will be challenging for treating human patients. Recently, luteolin (a compound isolated from the Bhutanese daisy, *Ajania nubigena*), bergapten and isomyristicin (both isolated from *Pleurospermum amabile*) have all been shown to have activity against *T. muris*, in each case associated with cuticular damage [21,22].

The lifecycle of *T. trichiura* commences with unembryonated eggs being shed in the faeces of infected individuals. These eggs embryonate in the soil, thus becoming infective, and are transmitted to their human host by consumption of contaminated food, water or soil. The eggs hatch in the large intestine, and infective larvae invade the mucosal epithelium. After a series of larval moults, adult worms develop and release eggs into the lumen of the caecum. Several life cycle stages of *Trichuris* have the potential as viable targets for small molecule drugs including immobilisation or killing of worms inside the host following infection; preventing egg embryonation in the external environment; and reducing the infectivity of embryonated eggs prior to ingestion (Fig 1). Thus in addition to post-infection eradication of existing infections, the targeting of eggs and embryonation may break the parasite life-cycle and thereby offer potential for infection control strategies that avoid costly and impractical pharmaceutical development (e.g. an environmental spray targeting eggs in environmental hotspots such as latrines).

*T. muris*, is widely used as a laboratory model for investigations on trichuriasis, and severe combined immune deficiency (SCID) mice have been used as a system for culture of *T. muris* and for *in vivo* testing of the capacity of anthelmintic drug candidates to clear infection in the absence of adaptive immunity [23]. Given the close similarity between *T. muris* and *T. trichiura* at both the genetic and proteomic level [24] we embarked on a programme of drug discovery searching for new small molecules active against *T. muris* with a view to generating anthelmintic leads. Here we present our discovery of the dihydrobenzoxazepinones, a new class of molecules active against *T. muris*. These molecules paralyse the adult parasite, and also reduce the ability of treated eggs to establish infection in the host *in vivo*.

#### **Methods**

#### Ethics statement

All animal experiments were approved by the University of Manchester Animal Welfare and Ethical Review Board and performed under the regulation of the United Kingdom Home Office Scientific Procedures Act (1986) and the Home Office project licence 70/8127.



**Fig 1. Schematic showing possible treatment strategies: post-infection treatment and breaking the** *Trichuris* **life cycle in the environment.** *Trichuris* infection could be prevented by using novel drugs to 1) target the prevention of egg embryonation in the external environment, 2) reduce the infectivity of embryonated eggs prior to ingestion, or 3) treat existing infections *in vivo* while worms are at larval and adult stages.

doi:10.1371/journal.pntd.0005359.g001

#### In vivo culture of Trichuris muris

Severe combined immune deficiency (SCID) mice were bred in the Biological Services Facility at the University of Manchester. Both male and female SCID mice were infected with a high dose of 200 infective embryonated *T. muris* eggs via oral gavage. At day 35 post-infection, groups of 5 mice were killed and their caecae and colons removed. Guts were opened longitudinally, washed with pre-warmed Roswell Park Memorial Institute (RPMI) 1640 media supplemented with penicillin (500U/ml) and streptomycin (500µg/ml) and adult *T. muris* worms gently removed using fine forceps.

#### Ex vivo T. muris adult maintenance for motility screen

Worms removed from SCID mice were maintained in RPMI-1640 media supplemented with penicillin (500U/ml) and streptomycin (500 $\mu$ g/ml) at approximately 37°C and studied on the same day. Using fine forceps, individual live worms were placed into 96 well plates containing 75 $\mu$ l of RPMI-1640 medium supplemented with penicillin (500U/ml) and streptomycin (500 $\mu$ g/ml) plus 1% final concentration of DMSO or compound dissolved in DMSO. In the primary screen the test compound concentration was 100 $\mu$ M. Plates were then incubated at 37°C, 5% CO<sub>2</sub>, and motility was analysed after 24 hours.

#### Automated motility assay

Worm movement was quantified using an automated motility assay. This system gives an estimate of worm movement in each well by thresholding and quantifying pixel intensity variance over time. An earlier version of this system has been previously described [25,26]. One adult *T. muris* worm was used per well, with one well per compound in each replicate of the primary screen. The primary screen was repeated each week for a period of four weeks giving four independent biological replicates from separate parasite batches for each compound. Dose-response curves were calculated with the four factor log-logistic model using the R package *drc* [27].

#### Small molecule compound library

The compound library used in this study (Chemistry Research Laboratory, University of Oxford) consists of over 50,000 drug-like small molecules, which has been assembled from a range of external and internal (proprietary) sources. A pilot screen was undertaken with a 480 compound subset of the main library, containing a variety of chemical scaffolds which were considered to be particularly amenable to further medicinal chemistry optimisation. This pilot screening set comprised more than twenty different mono- and polycyclic heteroaromatic scaffolds, each of which was substituted with a range of functional groups. Compounds were provided as 10mM solutions in DMSO and frozen at -20°C until required.

The primary screen was carried out at 100 $\mu$ M final concentration. Assay plates were assigned unique IDs and statistical analysis was performed without knowledge of compound identities using R scripts. Once hit wells were identified, well/plate positions were mapped to compound identities using DataWarrior [28]. Compounds active in the primary screen were retested in a blind secondary screen (n = 5, 100 $\mu$ M). Compounds identified as having encouraging activity from the DMSO solution samples were then resynthesized for confirmatory testing.

#### Chemical synthesis

Compounds were synthesised from commercially available starting materials, and fully characterised by Infrared (IR) Spectroscopy, Mass Spectrometry (ESI-MS, HRMS-EI) and Nuclear Magnetic Resonance (<sup>1</sup>H, <sup>13</sup>C NMR, HSQC, HMBC). Full experimental details and analytical data are provided in <u>S1 Supporting Information</u>.

#### In vivo establishment of infection

100 infective embryonated eggs were incubated with water, water plus DMSO vehicle, or test compounds at a final concentration of  $100\mu$ M for 14 days at room temperature. Eggs were then washed, re-suspended in water and 30 eggs given to a mouse via oral gavage. At day 15 post-infection, mice were culled and the number of worms present in the caecae and colon was counted. One data point was excluded: a water control individual mouse where there were technical difficulties with the oral gavage leading to the mouse regurgitating some of the *T. muris* eggs. It was ear marked at day of gavage and removed from the analysis.

We collected data using the same experimental design on different occasions, these are referred to as batches. Data were obtained from three batches (DMSO and water controls, batches [A,B,C]) or two batches (OX02983 [A,C], OX03153 [B,C]) with the total number of mice used, n, as follows: DMSO: 14, water: 16, OX02983: 10, OX03153: 12. The data were analysed by 2-way ANOVA (worms ~ treatment + batch + treatment \* batch). There was no significant interaction effect (P = 0.14) or effect of batch (P = 0.47). We therefore combined the data from the three batches. A one-way ANOVA (worms ~ treatment) showed a significant difference between groups (F(3,48) = 8.3, P< 0.0005). We then conducted a post-hoc test using Tukey's honest significant difference test.

#### Embryonation assay

To assess whether the dihydrobenzoxazepinone compounds could also prevent embryonation, 100 un-embryonated eggs were treated with OX02983 and OX03153 at  $100\mu$ M for 56 days in the dark at room temperature, then examined under an Olympus SD-ILK microscope for signs of embryonation.

#### Cytotoxicity testing

CMT-93 murine rectal epithelial cells were maintained in Dulbecco's Modified Eagle's Medium (DMEM) with 10% fetal calf serum (FCS) and 1% penicillin-streptomycin solution (10,000U/ml). Cells were plated in 96-well tissue culture plates at a density of  $2x10^2$  cells per well in a volume of 100µl, and cultured for 96 hours (37°C, 5% CO<sub>2</sub>) until reaching 30% confluence. The growth medium was removed and replaced with treatment media (DMEM, 1% FCS, 1% penicillin-streptomycin solution, 100µl assay volume) supplemented with vehicle alone (DMSO), test compounds or chlorpromazine, a positive control for cytotoxicity (final concentrations from 0 to 100µM). Maximum vehicle concentration for all compounds was 0.5% DMSO. Cells were cultured with the test compounds for 72 hours before cytotoxicity measurement.

The WST-8 assay was performed using the Cell Counting Kit– 8 (Sigma Aldrich # 96992).  $10\mu$ l of the CCK-8 solution was added to each well, plates were incubated for a further 2 hours, before absorbance was measured at 450nm using a microplate reader. This assay measures the ability of metabolically active cells to reduce a water-soluble formazan dye.

Following the WST-8 assay, the medium was removed, cells were washed with 100 $\mu$ l PBS and then 100 $\mu$ l of neutral red medium (DMEM, 1% FBS and 33 $\mu$ g/ml neutral red dye (Sigma Aldrich #N2889) were added to each well, followed by incubation for a further 2 hours. The neutral red medium was removed, cells washed with 100 $\mu$ l of PBS and 100 $\mu$ l of destain solution (50% ethanol, 49% distilled water and 1% glacial acetic acid) were added. Each assay plate was shaken rapidly for 10 minutes before absorbance was measured at 540nm using a microplate reader. This assay measures the ability of living cells to actively take up the neutral red dye.

Results were analysed using GraphPad Prism and fitted using a log-logistic model.

## Compound property analysis and preliminary Structure-Activity Relationships (SAR)

Analyses of compound properties and structure-activity relationships were performed using DataWarrior [28] and *rcdk* [29].

#### Results

#### Screening for molecules that paralyse ex vivo T. muris adults

The biology of *Trichuris* is comparatively poorly understood. In particular, as a clade I nematode it is only distantly related to the model nematode *Caenorhabditis elegans* [24], so we have limited information on potential targets for *Trichuris* control. We therefore decided to undertake a whole-organism phenotypic screen, searching for novel small molecules that paralyse or otherwise reduce the movement of adult *T. muris* parasites, screened *ex vivo*. Using an automated high-throughput motility analysis system, we screened 480 drug-like compounds from compound library DMSO solution samples in a pilot screen, with 4 biological replicates. 96 DMSO-only wells were used as negative controls. As shown in still frames selected from the raw movie data (Fig 2A), control worms swim vigorously in solution, but the most active compounds reduce movement to zero.

The results are summarised as a volcano plot (Fig 2B). Compounds that reduced movement and were highly reproducible (statistically significant) are shown in the top left-hand corner of this plot. To select hit compounds for future analysis we applied cut-offs of a movement reduction of at least  $2^{10}$ -fold in the automated motility assay and a significance value compared to the DMSO-only controls of P < 0.005 (Mann-Whitney test). 20 compounds were selected for further study. These compounds were blindly rescreened in the same adult motility assay (Fig 2C). 14/20 compounds were active in this assay (P < 0.05, Mann-Whitney test with Holm's correction for multiple comparisons). Interestingly, 13 of these active compounds from the screen contained the same core dihydrobenz[e][1,4]oxazepin-2(3H)-one structure (Fig 2D, hereafter referred to as dihydrobenzoxazepinones, or DHBs). No anthelmintic activity has previously been reported for this structural family and indeed it has not been implicated in any invertebrate control approach. Importantly, the DHB chemical class is both drug-like [30], and highly tractable for modification of the various structural motifs therein. This is critical as it provides a means to explore synthetically the prospect of optimising further both the intrinsic activity and the physicochemical and pharmacokinetic properties of the compound class.

Six compounds caused an increase in movement of at least 5-fold in the primary screen. The biological significance of this is not clear, although in the nematode *Caenorhabditis elegans*, treatments including dietary restriction lead to swimming hyperactivity [31]. However given our focus on anthelmintic discovery we focused on the compounds that reduced movement in our assay.

#### Elaboration of the dihydrobenzoxazepinone active family

Critical to the development of small molecule drugs is the establishment of the relationship between structure and activity. This allows iterative design, testing and improvement of key characteristics that determine the success of a compound such as potency, toxicity and pharmacokinetics.



**Fig 2. Hit compounds identified in a screen for molecules that paralyse whole** *T. muris* **parasites.** (A) Movie frames showing complete worm paralysis. A montage of frames for control (top row wells) and hit compound treated (bottom row wells) is displayed; each frame is 4 seconds apart (B) Volcano plot of primary screen (n = 4, Mann-Whitney-Wilcoxon test, compounds tested at 100µM) showing significant reduction in worm movement for 20 candidate compounds. (C) Of the 20 hit compounds from the primary screen that were selected for further study, 14 were active in the secondary screen (P < 0.05, Mann-Whitney-Wilcoxon test, corrected for multiple comparisons by Holm's method. \*\*\* = P < 0.005, \*\* = P < 0.01, \* = P < 0.05.) (D) 13 of the active compounds contain the same core dihydrobenzoxazepinone group. R<sub>1</sub> = Aryl or Cycloalkyl. R<sub>2</sub> = Aryl or Heteraryl.

doi:10.1371/journal.pntd.0005359.g002

We have undertaken a preliminary analysis by examining the activity in our primary assay of a set of 192 related dihydrobenzoxazepinones with the core structure shown in Fig 2D. Of these 20 are significantly active in the *T. muris* adult motility assay. Because the assay is fully automated and provides a movement measurement for each well, we have determined a quantitative activity estimate for all 192 dihydrobenzoxazepinones in this assay (Fig 3A), where activity is the minus log<sub>2</sub> movement reduction compared to DMSO-only controls when assayed at 100µM. This allows us to relate structure to activity. There is little correlation between molecular weight and activity (Fig 3B, Spearman's  $\rho = 0.14$ ) but a stronger relationship between predicted hydrophobicity and activity (Fig 3C, Spearman's  $\rho = 0.33$ ). It has previously been reported that compounds with higher aLogP are more likely to accumulate within the nematode *C. elegans*, probably reflecting a greater ability to cross a permeability barrier within the nematode cuticle [32]. Therefore, the higher hydrophobicity of the active dihydrobenzoxazepinones may reflect greater compound access into the nematode parasite.

Complete screening results and structures of the analysed dihydrobenzoxazepinones are presented in the <u>S1 Dataset</u>.

#### Resynthesis of hits

Following any screening campaign it is routine to triage and confirm activity of putative active compounds by sourcing and retesting authentic, unambiguously characterised solid samples [33]. This is important since DMSO solution samples can degrade over time, and this often leads to so-called 'false positive' hits. We established a flexible and general synthetic route to this family of dihydrobenzoxazepinones (Fig 4A). Five members were resynthesized in the first instance (Fig 4B), using the protocol outlined in the reaction scheme (see S1 Supporting Information for further details). This modular route has the advantage of being able to provide access to a much wider set of compounds for a future optimisation programme. Details of the resynthesized compounds are given in Table 1.

# Resynthesized dihydrobenzoxazepinones show dose-dependent activity against *T. muris* parasites

Having resynthesized selected members of the active dihydrobenzoxazepinone class, we wanted to confirm their activity in the *ex vivo* adult *T. muris* motility assay. We tested each



**Fig 3.** Elaboration of initial active compounds by characterising a diverse set of 192 dihydrobenzoxazepinones with the same core structure. (A) Screening and elaboration procedure. (B) Relationship between activity and molecular weight of each dihydrobenzoxazepinone. Activity is the minus log<sub>2</sub> movement reduction compared to DMSO-only controls when assayed at 100µM. (C) Relationship between activity and predicted hydrophobicity (cLogP) of each dihydrobenzoxazepinone.

doi:10.1371/journal.pntd.0005359.g003



Fig 4. Synthetic route to putative hit compounds. (A) i.  $R^1$ CHO (1.5 equiv.), AcOH (0.5 equiv.), NaBH(OAc)<sub>3</sub>, CH<sub>2</sub>Cl<sub>2</sub>, 0°C to RT, 2 days; ii. LiAlH<sub>4</sub> (1M in THF, 3.5 equiv.); iii. chloroacetyl chloride (2.0–4.0 equiv.), NEt<sub>3</sub> (2.0–4.4 equiv.), THF, 0°C to RT, 16 h; iv. 10N NaOH (aq.), RT, 2 h; v.  $R^2$ B(OH)<sub>2</sub> (1.3 equiv.), Pd(PPh<sub>3</sub>)<sub>4</sub> (5 mol%), 1.5M NaHCO<sub>3 (aq.)</sub> (3 equiv.), DMF, 150 °C (m.w.), 15 min. (B) Structures of resynthesized compounds.

doi:10.1371/journal.pntd.0005359.g004

Table 1. Structures of active dihydrobenzoxazepinones that were resynthesized, with selected calculated properties. RMM: relative molecular mass. #HBA: number of hydrogen bond acceptors. #HBD: number of hydrogen bond donors. tPSA: topological polar surface area. (Calculated using Data-Warrior [28]).

Compound	PubChem SID	R <sub>1</sub>	R <sub>2</sub>	RMM	cLogP	#HBA	#HBD	tPSA (Å <sup>2</sup> )
OX02983	318018055	4-Tolyl	7-(Pyrid-4-yl)	344	3.1	3	0	42
OX02993	318018058	Cyclohexyl	7-(Pyrid-3-yl)	336	2.9	3	0	42
OX03144	318018059	4-CF₃Ph	7-(Pyrid-3-yl)	398	3.6	6	0	42
OX03146	318018057	4-Tolyl	7-Br	346	3.6	2	0	30
OX03153	318018056	4-Tolyl	7-(Pyrid-3-yl)	344	3.1	3	0	42

doi:10.1371/journal.pntd.0005359.t001

NEGLECTED TROPICAL DISEASES



Fig 5. Resynthesized dihydrobenzoxazepinones show dose-dependent attenuation of *ex vivo T*. *muris* adult motility. Single adult worms were incubated for 24 hours in wells containing media plus compound. Motility was quantified using an automated phenotyping platform.  $EC_{50} \pm$  standard error shown in parentheses. n = 5, except OX03144 where n = 10. Curve fitted using the four parameter log-logistic model.

doi:10.1371/journal.pntd.0005359.g005

compound and determined the concentration-dependence of their activity between 1–200 $\mu$ M. These results are presented in Fig 5.

The EC<sub>50</sub> values of these compounds, around 25–50 $\mu$ M, are acceptable considering this assay screens the whole parasite where compound access and compound metabolism will limit the engagement of the molecules with the target. Encouragingly these EC<sub>50</sub> values are comparable with or exceed those for existing major anthelmintics in related *T. muris ex vivo* motility assays. Ivermectin is not active (maximum concentration tested 200 $\mu$ g/ml or 229 $\mu$ M) against *T. muris* L4 larvae [16] or against L1 larvae (maximum concentration tested 50 $\mu$ g/ml or 57 $\mu$ M) [14]. Mebendazole is not active against *T. muris* L1 larvae (maximum concentration tested 50 $\mu$ g/ml or 169 $\mu$ M) [14] but was active at 200 $\mu$ g/ml (678 $\mu$ M) though not 100 $\mu$ g/ml (339 $\mu$ M) against L3/L4 and adult *T. muris* [23]. Levamisole is active against *T. muris* adults in the automated assay utilised in this study (EC<sub>50</sub> = 8  $\mu$ M). Levamisole is effective at treating trichuriasis *in vivo*, although the single-dose cure rate, 9%, is low. This suggests that these early-stage dihydrobenzoxazepinone compounds already show promise for successful development as anti-*Trichuris* agents and supports future optimisation of structure and activity.

Structures and assay data for the five confirmed hit dihydrobenzoxazepinones have been submitted to the database PubChem (<u>https://pubchem.ncbi.nlm.nih.gov/</u>). Accession numbers for this data are shown in Table 1.

# Dihydrobenzoxazepinones are selective for the parasite with limited cytotoxicity to mouse gut epithelial cells

It was critical to ensure that our compounds showed selective activity against the parasite, as opposed to being generally toxic; as for example gut cytotoxicity may result in a limited therapeutic window. To address this, we assessed the resynthesized dihydrobenzoxazepinones for cytotoxicity in the mouse gut epithelial cell line CMT-93. We used two distinct assays, the WST-8 assay, which measures metabolic activity, and the neutral red accumulation assay, which measures the ability of living cells to actively take up the dye. Chlorpromazine was used as a positive control for cytotoxicity. The results of these experiments are shown in Table 2. Most of the dihydrobenzoxazepinones had no or weak cytotoxicity ( $EC_{50} > 100 \mu$ M). Interestingly, the most cytotoxic compound tested, OX03146, replaces the 7-pyridyl group with a smaller and more lipophilic 7-bromo substituent (Table 1). Along with the reduced polar surface area (tPSA), and likely improved membrane permeability [34], this might make the compound more promiscuous in binding to off-target proteins, thereby potentially explaining the increased cytotoxicity.

Taken together, these data suggest that at this early stage, our dihydrobenzoxazepinone hit compounds show adequate selectivity for the parasite target versus the murine/human host. Improving this selectivity will be one of the goals of the continuing medicinal chemistry program to progress the optimisation and development of this family of compounds.

## Treatment of eggs with dihydrobenzoxazepinones reduces their ability to establish infection *in vivo*

Developing a novel pharmaceutical for a neglected tropical disease such as trichuriasis is an economic and practical challenge. We wanted to consider whether an alternative strategy could be used to establish control of infection but with a potentially faster development process. As described in Fig 1, the lifecycle of both *T. muris* and *T. trichiura* involves a stage where unembryonated, uninfective eggs are released into the environment in the faeces. These eggs embryonate to become infective over a period of time and are then transmitted to further individuals via consumption of contaminated food or water. Therefore, interruption of the lifecycle by targeting of eggs in the environment is a potential alternative strategy to mass pharmaceutical treatment. We wanted to determine whether our compounds, when applied to eggs as if via an environmental treatment, could affect two key processes: establishment of infection and embryonation. We tested the first possibility by establishing an assay where embryonated *T. muris* eggs are soaked in the test compound, washed, and then applied to mice via oral gavage. Infectivity was quantified by culling the mice on day 15, and counting the number of worms that were hatched and able to establish infection in the mouse gut (Fig 6A). The results

Table	2. Dihydrobenzoxazepinones cause limited cytotoxicity against mouse epithelial cell lines demonstrating selectivity for the pathogen tar-
get. №	Iouse CMT-93 rectal epithelial cells were used for this assay. Maximum tested concentration was 100 µM. n = 8, error range (in parentheses) shows
95% c	onfidence interval. EC <sub>50</sub> values in the adult <i>Trichuris</i> paralysis assay (Fig 5) are shown for comparison.

Compound	EC <sub>50</sub> WST-8 assay (µM)	$EC_{50}$ Neutral red assay (µM)	$EC_{50}$ in adult $\textit{Trichuris}$ paralysis assay ( $\mu M$ )
Chlorpromazine (positive control)	9 (5–15)	10 (7–16)	n/a
OX02983	>100	>100	50
OX02993	>100	>100	52
OX03144	>100	60 (39–93)	26
OX03146	63 (37–111)	47 (31–71)	35
OX03153	>100	>100	57

doi:10.1371/journal.pntd.0005359.t002



**Fig 6.** *T. muris* eggs treated with dihydrobenzoxazepinone OX02983 are less infective *in vivo*. (A) Experimental scheme: embryonated eggs were soaked in compound, and then used to infect mice by oral gavage. At day 15 post-infection, mice were culled and worm burden assessed. (B) Treatment with OX02983 reduced the ability of embryonated eggs to establish infection *in vivo*. Blue bar indicates median worm burden. A one-way ANOVA (worms ~ treatment) showed a significant difference between treatment groups (F(3,48) = 8.3, P< 0.0005). Differences between groups were determined using a post-hoc Tukey HSD test (\*\* = P<0.005, \*\*\* = P<0.0001). n = 14 (DMSO), 16 (water), 10 (OX02983), 12 (OX03153). (C) Dihydrobenzoxazepinones do not act by blocking embryonation. Eggs were incubated with 100µM compounds or DMSO-alone control for 56 days. Embryonation was then quantified. No significant differences between groups were detected: one-way

ANOVA (embryonation ~ treatment, F(2,12) = 0.60, P = 0.57). Blue bar indicates median percentage embryonation.

doi:10.1371/journal.pntd.0005359.g006

are summarised in Fig 6B. Dihydrobenzoxazepinone OX02983 was able to reduce infectivity *in vivo* (Tukey's honest significant difference test, P < 0.005), by an estimated 54% (median DMSO control = 12.0 worms establishing infection, median OX02983 treated group = 5.5 worms). OX03153 was not active in this assay.

In contrast, the compounds were not able to block the earlier process of embryonation, where *T. muris* develops from a ball of cells into vermiform stage within the eggshell. This may reflect poor permeability of the compounds into the eggs, or indicate that they act later in development or interfere with the process of hatching or infection itself (Fig 6C).

These results support the potential use of dihydrobenzoxazepinones as a spray or other treatment that reduces the infectivity of *Trichuris* eggs within the environment, in particular at hotspots such as latrines. Such a use would however require improved potency to be achieved during future drug development.

#### Discussion

#### Current lack of effective treatments for trichuriasis

There are several distinct modes of action of current anthelmintics. Examples include benzimidazoles such as albendazole and mebendazole, which act by inhibiting microtubule synthesis [7], oxamniquine, which prevents nucleic acid synthesis [35], praziquantel, which is thought to act by increasing the permeability of schistosome cell membranes to calcium ions during infection [36], ivermectin inhibits parasite neurotransmission mainly via actions on Lglutamate gated chloride channels [37–40] and levamisole, oxantel and pyrantel are nicotinic receptor agonists [41]. However the limitations of current anthelmintic treatments are particularly pronounced in the context of regimes targeting *Trichuris*.

Mass Drug Administration (MDA) programs that aim to reduce and potentially eliminate soil-transmitted helminth infection in humans have predominantly relied on benzimidazole drugs alone, making this method of treatment vulnerable to the development of anthelmintic resistance [42]. Furthermore, single doses of benzimidazoles lead to low cure rates for infection with *T. trichiura*, only 28% and 36% for albendazole and mebendazole respectively [8]. Because of these low cure rates, modelling studies have concluded that MDA with benzimidazoles is unable to interrupt transmission of *T. trichiura* and thus achieve elimination in many settings [9].

Attempts have been made to boost the low single-dose cure rates of the benzimidazoles by combining them with other drugs. Studies conducted in school-aged children on Pemba Island, Tanzania, combined albendazole with a second anti-protozoal drug, nitazoxanide, which interferes with parasite anaerobic energy metabolism [43]. This combination treatment showed low cure rates against *T. trichiura* [44]. Similarly, drug trials conducted in Tanzania using combined treatments of albendazole plus ivermectin, show significant improvement in cure rates compared to albendazole or mebendazole alone, but these cure rates are also low (27.5%). Albendazole plus mebendazole was also used in the same study but also show low efficacy, as only 8.4% of test participants were cured [10]. Further multi-drug trials on Pemba Island Tanzania involving treatment with oxantel pamoate plus albendazole have resulted in higher cure (31% in one study and 68.5% in a second) and egg-reduction rates for *T. trichiura* infection when compared with standard therapy, however mild adverse effects were also reported [10,11]. These multi-anthelmintic drug treatments provide improved cure rates to *T*.

*trichiura* infection, when compared to the use of single anthelmintics alone, but risk unwanted side effects and are not appropriate for MDA. Recently a dose-ranging study measured the effectiveness of oxantel pamoate alone against *T. trichiura* in children and found, at the optimum dose, a cure rate of 60% with a good tolerability profile [12]. This drug is clearly the most promising current alternative to the benzimidazoles. However it is not a panacea with limited cure rates [45], still low compared to anti-hookworm MDA. Modelling also suggests that even improved cure rates with drug-combination MDA are insufficient to achieve elimination in higher transmission settings [9].

Despite the clear need for alternative anthelmintic drugs to achieve control or ideally eradication of diseases such as trichuriasis, the pipeline for development of such drugs is dry [11]. A particularly important reason for the poor efficacy of existing drugs for trichuriasis is that they were developed primarily for animal health indications, for the control of major agricultural pests such as *Haemonchus contortus*. *Trichuris*, being a clade I nematode is evolutionarily distant from most of these economically important parasites, so drug targets for the control of this species are likely to be divergent. We therefore decided to initiate a program of drug discovery that was targeted to *Trichuris* from the very start by screening for loss of motility in *T. muris* adults.

## Dihydrobenzoxazepinones, a new class of anthelmintics active on *T*. *muris*

Following a pilot high-throughput screen, subsequent hit triage and confirmatory resynthesis, we have identified the dihydrobenzoxazepinones as a new class of drug-like molecules with activity on the model mouse parasite, *T. muris.* EC<sub>50</sub> values of around 25–50 $\mu$ M for these early-stage, small molecules, activity data are comparable with the existing anthelmintic levamisole in the same assays. Encouragingly, cytotoxicity testing on murine gut epithelial cells showed that most compounds in this series had limited cytotoxicity, with EC<sub>50</sub> values > 100 $\mu$ M, the maximum concentration tested. Further toxicity testing in diverse *in vitro* and *in vivo* systems will be required during the development of this compound series. However these data indicate that the dihydrobenzoxazepinone (DHB) class have selectivity for the parasite, and are encouraging in the context of optimising this series further for post-infection treatment, which will require strict safety and host tolerance criteria.

Importantly, the DHB chemical class is both drug-like [30], and highly tractable for modification of the various structural motifs therein. New analogues can be designed to improve the broader molecular properties of the class, this being critical for modification of key parameters such as solubility, permeability and metabolism–all of which are essential for development of a new therapeutic agent. It is important to note that the term 'drug-like' as employed in the contemporary medicinal chemistry literature is usually directed towards orally delivered systemic agents which are targeting a central or peripherally located target enzyme or receptor. For our compounds, which are targeting the gastrointestinal located *Trichuris*, it would be desirable to have compounds with relatively *minimal* systemic exposure, and therefore the conventional 'drug-like' parameters which affect solubility and permeability, including lipophilicity need not adhere to the 'normal' Lipinski type values [46]. In addition, a relative metabolic lability may be desirable, such that any systemically absorbed compound would be rapidly cleared.

The desirable absorption and distribution properties of an anti-*Trichuris* agent are complicated by the burrowing of the whipworm into intestinal epithelial cells, which may provide some protection from drugs. There is limited understanding of this aspect of *Trichuris* biology, but a study has determined, after oral treatment of pigs infected with whipworm *T. suis*, the concentration of various benzimidazole drugs in host plasma and absorbed into the parasite [47]. The tight relationship between plasma and worm concentrations of oxfendazole led to the conclusion that this drug reaches the worm from the circulation via intestinal cells, whereas the limited relationship between plasma and worm concentrations of febendazole indicated direct absorption from the intestinal lumen. The role of the bacillary band, a specialised epidermal structure, in drug uptake is also thought to be important [48]. Quantifying the absorption route and kinetics of the DHB compounds will be required to generate a pharmacokinetic model and therefore critical to the further development of this series as an anti-*Trichuris* agent.

No anthelmintic activity has previously been reported for the DHB structural family and indeed it has not been implicated in any invertebrate control approach. Although we have not as yet identified the molecular target for this class of molecules, the dihydrobenzoxazepinones are a novel family of anthelmintics specifically identified for their activity on *Trichuris*.

#### Developmental route to a new oral drug for trichuriasis

The developmental process leading to the identification of a new drug is long, expensive, and prone to failure. The hit to lead stage of this process takes molecules identified in highthroughput screens and iteratively improves their activity in disease-relevant assays as well as building in properties important for drug action, particularly related to drug metabolism and pharmacokinetics (DMPK). Recent experience in drug development of neglected diseases has led to the establishment of criteria for the hit-to-lead process [49,50]. The DHB compounds meet all of the criteria for hit selection for infectious diseases of the developing world, such as confirmed activity with resynthesized compound, sigmoidal dose-dependence, tractable chemotype that passes drug-like filters, and a selectivity window for cytotoxicity in a mammalian cell line [49]. These compounds also meet some (but not all) of the criteria for lead compounds, such as establishment of a tractable route for synthesis that is amenable to rapid and diverse series expansion, and suitable drug-likeness. The *in vitro* activity of the compounds in motility assays ( $26-57\mu$ M, equivalent to  $10-20\mu$ g/ml) also reaches the lead compound criterion (10µg/ml) for motility inhibition against the related microfilarial nematode Brugia malayi [50]. The largest gap in our understanding of the DHB compounds is related to DMPK. As discussed in the previous section, understanding these processes are critical to the development of orally active small molecules. If the DMPK properties of the DHB compounds can be determined and where necessary improved, it is possible that this chemotype will achieve the final milestone for lead compounds against helminth disease, namely a substantial reduction in worm burden in in vivo tests.

## Reduction of infectivity of eggs offers prospect of environmental treatment for trichuriasis

An exciting and complementary use of the novel anthelminitics is to use them in the environment to decrease the risk of infection by targeting the egg stage of the *T. trichiura* life cycle. This approach is particularly attractive as *T. trichiura* eggs are highly resistant to extreme temperature changes and ultraviolet radiation and thus remain viable in the environment for many years [51]. Dihydrobenzoxazepinone OX02983 was found to significantly reduce infectivity of eggs *in vivo* compared to eggs that had been treated under the same conditions with water or DMSO alone (Fig 6B). In contrast, there were no differences between the percentage of embryonation seen between dihydrobenzoxazepinone treated eggs and eggs treated with water or DMSO alone. This suggests that none of the active compounds were able to prevent embryonic development.

Further investigation is required to identify the mode of action of this compound on embryonated eggs. For example, the compound may reduce L1 viability within the egg, or alter the eggshell structure, preventing larval hatching inside the host. Chemical control of egg hatching would be particularly beneficial in breaking the *T. trichiura* life cycle in the field before the parasite can infect or re-infect the host. This mechanism of action would require environmentally stable and non-toxic compositions.

We envisage that, if an increase in potency is achieved during the drug development process, dihydrobenzoxazepinones could potentially be developed into a therapeutic agent that acts in the environment, perhaps via a spray utilised at sites of high-parasite density such as communal latrines, to interrupt the parasite lifecycle. This could allow for wider deployment in environments where mass drug administration is challenging. Consideration of environmental toxicity would be critical in this context. A further possibility is that mass drug administration with a compound with both anti-worm and anti-egg infectivity activity might, in addition to worm killing/expulsion, achieve a reduction of egg infectivity in the faeces, helping to break the parasite lifecycle.

#### Possible animal health applications

This study was initiated based on the urgent need for new treatments for the neglected human disease trichuriasis. However the economic realities of the cost of drug development may mean that the financial impetus behind the development of new drugs for *T. trichiura* comes from the animal health industry via control of *Trichuris* species that infect agricultural and companion animals. The related species *Trichuris suis* and *Trichuris vulpis* infect pigs and dogs respectively, with *T. suis* infection causing economically-costly reduction in pig weight-gain [52–54]. Interestingly increasing outdoor rearing of pigs due to animal welfare considerations has resulted in accumulation of long-lived *T. suis* eggs in the soil, causing difficulties with controlling infection [52]. Therefore an environmental egg-targeted treatment that interrupts the parasite lifecycle may be particularly useful in this context.

#### Conclusions

This study has identified a new family of molecules active against the important parasite *Trichuris.* We have found 20 active family members, from a total of 192 dihydrobenzoxazepinones tested, the activity being confirmed through resynthesis. This has allowed us to start to understand the molecular features that are important for activity against the parasite. The dihydrobenzoxazepinone compounds that we have identified are drug-like and ideal for further iterative optimisation and improvement as part of the drug development process. One member of the dihydrobenzoxazepinone class, OX02983, was also effective at reducing the ability of eggs to establish infection *in vivo*, pointing the way to a potential environmental treatment for trichuriasis. The successful identification of anthelmintic compounds using our automated motility phenotyping platform supports the utility of this approach, and could potentially be used with a variety of animal and human nematodes and other parasites.

#### **Supporting information**

S1 Supporting Information. Methodological detail and supporting spectra for the resynthesized dihydrobenzoxazepinone compounds. (DOCX)

S1 Dataset. File containing structure and assay information on screened dihydrobenzoxazepinones. (XLSX)

#### Acknowledgments

We thank A.E. Brown and Dr. S.D. Buckingham for helpful discussions.

#### **Author Contributions**

Conceptualization: FAP EAM NJW GMW KJE AJR DBS.

Formal analysis: FAP EAM NJW RF NHC GMW.

Funding acquisition: KJE AJR DBS.

Investigation: FAP EAM NJW CJRB RF NHC BM DJCS GMW KJE.

Methodology: FAP EAM NJW RF NHC GMW KJE AJR DBS.

Project administration: KJE AJR DBS.

Software: FAP.

Supervision: KJE AJR DBS.

Visualization: FAP EAM NJW GMW.

Writing - original draft: FAP NJW GMW DBS.

Writing - review & editing: FAP EAM NJW GMW KJE AJR DBS.

#### References

- Pullan RL, Smith JL, Jasrasaria R, Brooker SJ. Global numbers of infection and disease burden of soil transmitted helminth infections in 2010. Parasit Vectors. 2014; 7: 37. doi: 10.1186/1756-3305-7-37 PMID: 24447578
- 2. Harris NL. Advances in helminth immunology: optimism for future vaccine design? Trends Parasitol. 2011; 27: 288–293. doi: 10.1016/j.pt.2011.03.010 PMID: 21531626
- Vercruysse J, Albonico M, Behnke JM, Kotze AC, Prichard RK, McCarthy JS, et al. Is anthelmintic resistance a concern for the control of human soil-transmitted helminths? Int J Parasitol Drugs Drug Resist. 2011; 1: 14–27. doi: 10.1016/j.ijpddr.2011.09.002 PMID: 24533260
- Vercruysse J, Levecke B, Prichard R. Human soil-transmitted helminths: implications of mass drug administration. Curr Opin Infect Dis. 2012; 25: 703–708. doi: 10.1097/QCO.0b013e328358993a PMID: 22964945
- Diawara A, Schwenkenbecher JM, Kaplan RM, Prichard RK. Molecular and Biological Diagnostic Tests for Monitoring Benzimidazole Resistance in Human Soil-Transmitted Helminths. Am J Trop Med Hyg. 2013; 88: 1052–1061. doi: 10.4269/ajtmh.12-0484 PMID: 23458960
- Levecke B, Montresor A, Albonico M, Ame SM, Behnke JM, Bethony JM, et al. Assessment of Anthelmintic Efficacy of Mebendazole in School Children in Six Countries Where Soil-Transmitted Helminths Are Endemic. PLOS Negl Trop Dis. 2014; 8: e3204. doi: <u>10.1371/journal.pntd.0003204</u> PMID: 25299391
- Theodorides VJ, Gyurik RJ, Kingsbury WD, Parish RC. Anthelmintic activity of albendazole against liver flukes, tapeworms., lung and gastrointestinal roundworms. Experientia. 1976; 32: 702–703. PMID: 950011
- Keiser J, Utzinger J. Efficacy of current drugs against soil-transmitted helminth infections: Systematic review and meta-analysis. JAMA. 2008; 299: 1937–1948. doi: 10.1001/jama.299.16.1937 PMID: 18430913
- Turner HC, Truscott JE, Bettis AA, Hollingsworth TD, Brooker SJ, Anderson RM. Analysis of the population-level impact of co-administering ivermectin with albendazole or mebendazole for the control and elimination of *Trichuris trichiura*. Parasite Epidemiol Control. 2016; 1: 177–187. doi: 10.1016/j.parepi. 2016.02.004 PMID: 27430028
- **10.** Speich B, Ali SM, Ame SM, Bogoch II, Alles R, Huwyler J, et al. Efficacy and safety of albendazole plus ivermectin, albendazole plus mebendazole, albendazole plus oxantel pamoate, and mebendazole alone against *Trichuris trichiura* and concomitant soil-transmitted helminth infections: a four-arm,

randomised controlled trial. Lancet Infect Dis. 2015; 15: 277–284. doi: 10.1016/S1473-3099(14)71050-3 PMID: 25589326

- Speich B, Ame SM, Ali SM, Alles R, Huwyler J, Hattendorf J, et al. Oxantel Pamoate–Albendazole for Trichuris trichiura Infection. N Engl J Med. 2014; 370: 610–620. doi: 10.1056/NEJMoa1301956 PMID: 24521107
- Moser W, Ali SM, Ame SM, Speich B, Puchkov M, Huwyler J, et al. Efficacy and safety of oxantel pamoate in school-aged children infected with *Trichuris trichiura* on Pemba Island, Tanzania: a parallel, randomised, controlled, dose-ranging study. Lancet Infect Dis. 2016; 16: 53–60. doi: 10.1016/S1473-3099 (15)00271-6 PMID: 26388169
- Olsen A. Efficacy and safety of drug combinations in the treatment of schistosomiasis, soil-transmitted helminthiasis, lymphatic filariasis and onchocerciasis. Trans R Soc Trop Med Hyg. 2007; 101: 747–758. doi: 10.1016/j.trstmh.2007.03.006 PMID: 17481681
- Wimmersberger D, Tritten L, Keiser J. Development of an *in vitro* drug sensitivity assay for *Trichuris muris* first-stage larvae. Parasit Vectors. 2013; 6: 42. doi: 10.1186/1756-3305-6-42 PMID: 23433224
- Tritten L, Silbereisen A, Keiser J. In vitro and in vivo efficacy of Monepantel (AAD 1566) against laboratory models of human intestinal nematode infections. PLoS Negl Trop Dis. 2011; 5: e1457. doi: 10. 1371/journal.pntd.0001457 PMID: 22216366
- Silbereisen A, Tritten L, Keiser J. Exploration of novel in vitro assays to study drugs against *Trichuris* spp. J Microbiol Methods. 2011; 87: 169–175. doi: 10.1016/j.mimet.2011.08.009 PMID: 21889548
- Hansen TVA, Nejsum P, Friis C, Olsen A, Thamsborg SM. *Trichuris suis* and *Oesophagostomum dentatum* Show Different Sensitivity and Accumulation of Fenbendazole, Albendazole and Levamisole In Vitro. PLOS Negl Trop Dis. 2014; 8: e2752. doi: 10.1371/journal.pntd.0002752 PMID: 24699263
- Koné WM, Vargas M, Keiser J. Anthelmintic activity of medicinal plants used in Côte d'Ivoire for treating parasitic diseases. Parasitol Res. 2011; 110: 2351–2362. doi: <u>10.1007/s00436-011-2771-z</u> PMID: 22200959
- Stepek G, Lowe AE, Buttle DJ, Duce IR, Behnke JM. In vitro and in vivo anthelmintic efficacy of plant cysteine proteinases against the rodent gastrointestinal nematode, *Trichuris muris*. Parasitology. 2006; 132: 681–689. doi: 10.1017/S003118200500973X PMID: 16448585
- Levecke B, Buttle DJ, Behnke JM, Duce IR, Vercruysse J. Cysteine proteinases from papaya (Carica papaya) in the treatment of experimental *Trichuris suis* infection in pigs: two randomized controlled trials. Parasit Vectors. 2014; 7: 255. doi: 10.1186/1756-3305-7-255 PMID: 24886388
- Wangchuk P, Pearson MS, Giacomin PR, Becker L, Sotillo J, Pickering D, et al. Compounds Derived from the Bhutanese Daisy, *Ajania nubigena*, Demonstrate Dual Anthelmintic Activity against *Schisto-soma mansoni* and *Trichuris muris*. PLOS Negl Trop Dis. 2016; 10: e0004908. doi: <u>10.1371/journal.pntd.0004908</u> PMID: 27490394
- Wangchuk P, Giacomin PR, Pearson MS, Smout MJ, Loukas A. Identification of lead chemotherapeutic agents from medicinal plants against blood flukes and whipworms. Sci Rep. 2016; 6: 32101. doi: <u>10.</u> 1038/srep32101 PMID: 27572696
- 23. Hurst RJ, Hopwood T, Gallagher AL, Partridge FA, Burgis T, Sattelle DB, et al. An antagonist of the retinoid X receptor reduces the viability of *Trichuris muris* in vitro. BMC Infect Dis. 2014; 14: 520. doi: <u>10.</u> <u>1186/1471-2334-14-520 PMID: 25261877</u>
- Foth BJ, Tsai IJ, Reid AJ, Bancroft AJ, Nichol S, Tracey A, et al. Whipworm genome and dual-species transcriptome analyses provide molecular insights into an intimate host-parasite interaction. Nat Genet. 2014; 46: 693–700. doi: 10.1038/ng.3010 PMID: 24929830
- Buckingham SD, Sattelle DB. Fast, automated measurement of nematode swimming (thrashing) without morphometry. BMC Neurosci. 2009; 10: 84. doi: 10.1186/1471-2202-10-84 PMID: 19619274
- 26. Buckingham SD, Partridge FA, Sattelle DB. Automated, high-throughput, motility analysis in *Caenor-habditis elegans* and parasitic nematodes: Applications in the search for new anthelmintics. Int J Parasitol Drugs Drug Resist. 2014; 4: 226–232. doi: 10.1016/j.ijpddr.2014.10.004 PMID: 25516833
- Ritz C, Baty F, Streibig JC, Gerhard D. Dose-Response Analysis Using R. PLOS ONE. 2015; 10: e0146021. doi: 10.1371/journal.pone.0146021 PMID: 26717316
- Sander T, Freyss J, von Korff M, Rufener C. DataWarrior: An Open-Source Program For Chemistry Aware Data Visualization And Analysis. J Chem Inf Model. 2015; 55: 460–473. doi: 10.1021/ci500588j PMID: 25558886
- 29. Guha R. Chemical Informatics Functionality in R. J Stat Softw. 2007; 18.
- Bickerton GR, Paolini GV, Besnard J, Muresan S, Hopkins AL. Quantifying the chemical beauty of drugs. Nat Chem. 2012; 4: 90–98. doi: 10.1038/nchem.1243 PMID: 22270643

- Luersen K, Faust U, Gottschling D-C, Doring F. Gait-specific adaptation of locomotor activity in response to dietary restriction in *Caenorhabditis elegans*. J Exp Biol. 2014; 217: 2480–2488. doi: 10. 1242/jeb.099382 PMID: 24803455
- Burns AR, Wallace IM, Wildenhain J, Tyers M, Giaever G, Bader GD, et al. A predictive model for drug bioaccumulation and bioactivity in *Caenorhabditis elegans*. Nat Chem Biol. 2010; 6: 549–557. doi: <u>10</u>. <u>1038/nchembio.380 PMID</u>: 20512140
- Dahlin JL, Walters MA. The essential roles of chemistry in high-throughput screening triage. Future Med Chem. 2014; 6: 1265–1290. doi: 10.4155/fmc.14.60 PMID: 25163000
- Artursson P, Bergström CAS. Intestinal Absorption: The Role of Polar Surface Area. In: Waterbeemd H van de, Lennernäs H, Artursson P, editors. Drug Bioavailability. Wiley-VCH Verlag GmbH & Co. KGaA; 2003. pp. 339–357. Available: http://onlinelibrary.wiley.com/
- Cioli D, Pica-Mattoccia L, Archer S. Antischistosomal drugs: Past, present . . . and future? Pharmacol Ther. 1995; 68: 35–85. PMID: 8604437
- Doenhoff MJ, Cioli D, Utzinger J. Praziquantel: mechanisms of action, resistance and new derivatives for schistosomiasis. Curr Opin Infect Dis. 2008; 21: 659–667. doi: <u>10.1097/QCO.0b013e328318978f</u> PMID: 18978535
- Moreno Y, Nabhan JF, Solomon J, Mackenzie CD, Geary TG. Ivermectin disrupts the function of the excretory-secretory apparatus in microfilariae of *Brugia malayi*. Proc Natl Acad Sci. 2010; 107: 20120– 20125. doi: 10.1073/pnas.1011983107 PMID: 21041637
- Geary TG, Sims SM, Thomas EM, Vanover L, Davis JP, Winterrowd CA, et al. *Haemonchus contortus*: Ivermectin-Induced Paralysis of the Pharynx. Exp Parasitol. 1993; 77: 88–96. doi: <u>10.1006/expr.1993</u>. 1064 PMID: 8344410
- Kass IS, Wang CC, Walrond JP, Stretton AO. Avermectin B1a, a paralyzing anthelmintic that affects interneurons and inhibitory motoneurons in *Ascaris*. Proc Natl Acad Sci U S A. 1980; 77: 6211–6215. PMID: 6255481
- Raymond V, Sattelle DB. Novel animal-health drug targets from ligand-gated chloride channels. Nat Rev Drug Discov. 2002; 1: 427–436. doi: 10.1038/nrd821 PMID: 12119744
- 41. Williamson SM, Robertson AP, Brown L, Williams T, Woods DJ, Martin RJ, et al. The Nicotinic Acetylcholine Receptors of the Parasitic Nematode Ascaris suum: Formation of Two Distinct Drug Targets by Varying the Relative Expression Levels of Two Subunits. PLOS Pathog. 2009; 5: e1000517. doi: 10. 1371/journal.ppat.1000517 PMID: 19609360
- 42. Soukhathammavong PA, Sayasone S, Phongluxa K, Xayaseng V, Utzinger J, Vounatsou P, et al. Low efficacy of single-dose albendazole and mebendazole against hookworm and effect on concomitant helminth infection in Lao PDR. PLoS Negl Trop Dis. 2012; 6: e1417. doi: 10.1371/journal.pntd.0001417 PMID: 22235353
- Anderson VR, Curran MP. Nitazoxanide: a review of its use in the treatment of gastrointestinal infections. Drugs. 2007; 67: 1947–1967. PMID: 17722965
- 44. Speich B, Ame SM, Ali SM, Alles R, Hattendorf J, Utzinger J, et al. Efficacy and Safety of Nitazoxanide, Albendazole, and Nitazoxanide-Albendazole against *Trichuris trichiura* Infection: A Randomized Controlled Trial. PLOS Negl Trop Dis. 2012; 6: e1685. doi: 10.1371/journal.pntd.0001685 PMID: 22679525
- Maheu-Giroux M. Oxantel pamoate and deworming programmes. Lancet Infect Dis. 2016; 16: 5–6. doi: 10.1016/S1473-3099(15)00305-9 PMID: 26388170
- Lipinski CA, Lombardo F, Dominy BW, Feeney PJ. Experimental and computational approaches to estimate solubility and permeability in drug discovery and development settings. Adv Drug Deliv Rev. 1997; 23: 3–25.
- Hansen TVA, Friis C, Nejsum P, Olsen A, Thamsborg SM. Uptake of benzimidazoles by Trichuris suis in vivo in pigs. Int J Parasitol Drugs Drug Resist. 2014; 4: 112–117. doi: 10.1016/j.ijpddr.2014.03.003 PMID: 25057460
- Hansen TVA, Hansen M, Nejsum P, Mejer H, Denwood M, Thamsborg SM. Glucose Absorption by the Bacillary Band of Trichuris muris. PLoS Negl Trop Dis. 2016; 10: e0004971. doi: <u>10.1371/journal.pntd</u>. 0004971 PMID: 27588682
- Katsuno K, Burrows JN, Duncan K, van Huijsduijnen RH, Kaneko T, Kita K, et al. Hit and lead criteria in drug discovery for infectious diseases of the developing world. Nat Rev Drug Discov. 2015; 14: 751– 758. doi: 10.1038/nrd4683 PMID: 26435527
- Nwaka S, Ramirez B, Brun R, Maes L, Douglas F, Ridley R. Advancing Drug Innovation for Neglected Diseases—Criteria for Lead Progression. PLoS Negl Trop Dis. 2009; 3: e440. doi: 10.1371/journal. pntd.0000440 PMID: 19707561

- Maya C, Ortiz M, Jiménez B. Viability of Ascaris and other helminth genera non larval eggs in different conditions of temperature, lime (pH) and humidity. Water Sci Technol J Int Assoc Water Pollut Res. 2010; 62: 2616–2624.
- Roepstorff A, Mejer H, Nejsum P, Thamsborg SM. Helminth parasites in pigs: New challenges in pig production and current research highlights. Vet Parasitol. 2011; 180: 72–81. doi: 10.1016/j.vetpar.2011. 05.029 PMID: 21684689
- Hale OM, Stewart TB. Influence of an Experimental Infection of on Performance of Pigs. J Anim Sci. 1979; 49: 1000. PMID: 541282
- 54. Traversa D. Are we paying too much attention to cardio-pulmonary nematodes and neglecting old-fashioned worms like *Trichuris vulpis*? Parasit Vectors. 2011; 4: 32. doi: <u>10.1186/1756-3305-4-32</u> PMID: 21385441